Author's response to reviews

Title: Thoracic Epidural analgesia on Oxygenation and pulmonary Shunt Fraction during One-lung Ventilation: A meta-Analysis

Authors:

Xiao-Qian Li (shirley037305@hotmail.com)
Wen-Fei Tan (winfieldtan@hotmail.com)
Jun Wang (wjun2000@hotmail.com)
Bo Fang (drunk0630@aliyun.com)
Hong Ma (mahong5466@yahoo.com)

Version: 6
Date: 12 March 2015

Author's response to reviews: see over
Dear Editor:

Thank you very much for raising professional concerns and asking us to revise our subscription. After assessing a large amount of literature and obtaining recent experiments data, we provide a point-by-point response to each reviewer. Please do not hesitate to contact us if you have any questions or find anywhere inappropriate. We will carry out further research and make corrections accordingly. Thank you again and we are looking forward to being accepted by this journal.

Yours sincerely,

Hong Ma Ph. D

Email: mahong5466@yahoo.com
Address: No. 155 Nanjing North Street, Shenyang, Liaoning, China, 110001
Telephone number: +86 24 8328 3157
Facsimile number: +86 24 8328 2997

Replies to Editor

The manuscript needs a complete and thorough linguistic revision.

Answer: Thank you for the question raised by reviewer. We have modified the corresponding sections mentioned by the reviewer, according to the studies published in BMC Anesthesiology as well as the help of naïve English speaker.